Author's response to reviews

Title: Surface TRAIL decoy receptor-4 expression is correlated with TRAIL resistance in MCF7 breast cancer cells

Authors:

Ahter D. Sanlioglu (ahter@akdeniz.edu.tr)
Ercument Dirice (ercument@akdeniz.edu.tr)
Cigdem Aydin (cigdemaydin@akdeniz.edu.tr)
Nuray Erin (nerin@akdeniz.edu.tr)
Sadi Koksoy (sadikoksoy@akdeniz.edu.tr)
Salih Sanlioglu (sanlioglu@akdeniz.edu.tr)

Version: 4 Date: 31 March 2005

Author's response to reviews:

Dear Editor, I have included one more essential assay (a DCR2 siRNA approach) being the final assay to prove our point in our manuscript. This assay simply proves that TRAIL-R4 modulates TRAIL sensitivity in MCF7 breast cancer cells. We hope that this addition will expedite the Editorial decision on our manuscript. Sincerely